Author’s response to reviews

Title: Protective effect of Shenmai injection on doxorubicin-induced cardiotoxicity via regulation of inflammatory mediators

Authors:

Sheng Zhang (zs616568@163.com)
Zhen-Qiang You (youzq1979@163.com)
Lin Yang (luckbuff@163.com)
Li-li Li (15205807862@163.com)
You-Ping Wu (wuyouping@zju.edu.cn)
Li-qiang Gu (xiaoguddy@163.com)
Yan-fei Xin (xinyanfei@gmail.com)

Version: 6 Date: 03 Aug 2019

Author’s response to reviews:

Dear Editors and Reviewers,

Thank you for your letter and for the reviewers’ comments and potential accepting our manuscript entitled “Protective effect of Shenmai injection on doxorubicin-induced cardiotoxicity via regulation of inflammatory mediators” (BCAM-D-18-01258R1). We had studied the comments carefully and made correction which we hope meet with approval. Pointe-by-point responses to the editor are listed below this letter.

We hope the revised manuscript is now conformed to publication in your journal.

Best regards,

Yours sincerely,
Yanfei Xin.

Responds to the editor’ comments
1. Comments (C):
   Please clarify your anesthesia method including dosages. Please try to be as detailed as possible.

Reply (R):

Thank you for your suggestion. We have briefly described the euthanasia/sacrifice methods of
experimental animals in the method section (page 5, line 8). Now we re-describe this process in detail. Now we are writing the detailed method of anesthesia and execution of animals separately in the Method section “Animal (page 4, line 27)”.

2. Comments (C):
It has come to our attention that throughout your manuscript there is significant overlap with previously published works, in particular:
This overlap exists throughout your manuscript…
Reply (R):
We used the online check system to check the potential overlap content and carefully modify the statements that may be repeated throughout the article. We sincerely hope that the revised article content will be in line with the independent creation.

3. Comments (C):
Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Reply (R):
Thank you. We complied with this submission process.